Author's response to reviews

Title: Acute oxygenation changes on ischemic foot of a novel intermittent pneumatic compression device and of an existing sequential device in severe peripheral arterial disease.

Authors:

Fabio Manfredini (mdf@unife.it)
Anna Maria Malagoni (mlgnmr@unife.it)
Michele Felisatti (flsmhl@unife.it)
Simona Mandini (mndsmn@unife.it)
Nicola Lamberti (lmbncl@unife.it)
Roberto Manfredini (mfr@unife.it)
Francesco Mascoli (msf@unife.it)
Nino Basaglia (n.basaglia@ospfe.it)
Paolo Zamboni (zambo@unife.it)

Version: 2
Date: 19 November 2013

Author's response to reviews: see over
Dear Editor,

After a contact with the Editorial Office, a second version of the manuscript has been submitted with a few changes to the formatting, as follows:

1) The word count in the title has been reduced.
1) The word Introduction was changed with Background.
2) The words Subjects and Methods were changed with Methods.
3) A list of abbreviations has been added.
4) The section Figure Legends was moved at the end of the document.
5) The word count in the figure titles has been reduced.

We apologise for the formatting mistakes of the previous manuscript.

Kind regards

Anna Maria Malagoni

Address for correspondence:

Anna Maria Malagoni
Program Pathophysiology of Vascular Peripheral System and Day Surgery, S. Anna Hospital
University of Ferrara
Via Aldo Moro 8
44124 Ferrara-Loc. Cona, Italy
Tel +39 0532 239783
Fax +39 0532 237443
E-mail: mlgnmr@unife.it